Author's response to reviews

Title: Changes in abdominal obesity in university students stratified by Body Mass Index Changes in body fat in university students

Authors:

Marco Cossio-Bolaños (mcossio1972@hotmail.com)
Catalina Vilchez-Avaca (chaningc1@yahoo.com)
Victor Contreras-Mellado (rpcafd@gmail.com)
Cynthia Lee-Andruske (pesquisadores2010@gmail.com)
Rossana Gomez-Campos (rossaunicamp@gmail.com)

Version: 2  Date: 22 March 2015

Author's response to reviews: see over
Editor-in-Chief
Dear Editor,

Please find enclosed the manuscript entitled “Changes in abdominal obesity in university students stratified by Body Mass Index” by Cossio-Bolaños, Vilchez-Avaca, Contreras-Mellado, Lee-Andruske and Gómez-Campos. Find also enclosed a set of figures and tables. The authors state that this manuscript represents an original work, which has not been submitted elsewhere.

The article focuses on aspects of public health in university, determining the trend of change in body fat of university students in intervals of years. Allowing to know some of the factors determining the increase in the prevalence of overweight and obesity in recent years. Finally, provides a starting point for creating health policies in a given population.

This manuscript is one of a kind and all authors have contributed substantially to the manuscript and approved the final submission. I look forward to hearing from you at your earliest convenience.

Yours sincerely,
Marco Cossio Bolaños
(corresponding author)
Department of Physical Education
Catholic University of Maule
Talca
Chile
e-mail: mcossio1972@hotmail.com